UCSF UC San Francisco Previously Published Works

Title

Anti-ganglioside antibodies in patients with Zika virus infection-associated Guillain-Barré Syndrome in Brazil

Permalink https://escholarship.org/uc/item/61k051ft

Journal PLOS Neglected Tropical Diseases, 13(9)

ISSN

1935-2727

Authors

Rivera-Correa, Juan de Siqueira, Isadora Cristina Mota, Sabrina <u>et al.</u>

Publication Date

2019

DOI

10.1371/journal.pntd.0007695

Copyright Information

This work is made available under the terms of a Creative Commons Attribution License, available at <u>https://creativecommons.org/licenses/by/4.0/</u>

Peer reviewed



GOPEN ACCESS

Citation: Rivera-Correa J, de Siqueira IC, Mota S, do Rosário MS, Pereira de Jesus PA, Alcantara LCJ, et al. (2019) Anti-ganglioside antibodies in patients with Zika virus infection-associated Guillain-Barré Syndrome in Brazil. PLoS Negl Trop Dis 13(9): e0007695. https://doi.org/10.1371/journal.pntd.0007695

Editor: Rebecca C. Christofferson, Louisiana State University, UNITED STATES

Received: March 18, 2019

Accepted: August 7, 2019

Published: September 17, 2019

Copyright: © 2019 Rivera-Correa et al. This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: All relevant data are within the manuscript and its Supporting Information files

Funding: AR and JRC received funding for this project from New York University School of Medicine Internal Funds. The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

Competing interests: The authors have declared that no competing interests exist.

RESEARCH ARTICLE

Anti-ganglioside antibodies in patients with Zika virus infection-associated Guillain-Barré Syndrome in Brazil

Juan Rivera-Correa¹, Isadora Cristina de Siqueira², Sabrina Mota¹, Mateus Santana do Rosário³, Pedro Antônio Pereira de Jesus⁴, Luiz Carlos Junior Alcantara⁵, Joel D. Ernst^{1,6}, Ana Rodriguez¹*

 New York University School of Medicine, Dept. of Microbiology, New York, New York, United States of America, 2 Instituto Gonçalo Moniz, Fundação Oswaldo Cruz, Ministério da Saúde, Salvador, Brazil,
Hospital Santa Izabel, Salvador, Brazil, 4 Hospital Geral Roberto Santos, Secretaria Estadual da Saúde da Bahia, Salvador, Brazil, 5 Laboratório de Flavivírus—Instituto Oswaldo Cruz—Fundação Oswaldo Cruz, Rio de Janeiro, Brazil, 6 Division of Experimental Medicine, University of California, San Francisco, San Francisco, CA, United States of America

* ana.rodriguez@nyumc.org

Abstract

Zika virus infection is associated with the development of Guillain-Barré syndrome (GBS), a neurological autoimmune disorder caused by immune recognition of gangliosides and other components at nerve membranes. Using a high-throughput ELISA, we have analyzed the anti-glycolipid antibody profile, including gangliosides, of plasma samples from patients with Zika infections associated or not with GBS in Salvador, Brazil. We have observed that Zika patients that develop GBS present higher levels of anti-ganglioside antibodies when compared to Zika patients without GBS. We also observed that a broad repertoire of gangliosides was targeted by both IgM and IgG anti-self antibodies in these patients. Since Zika virus infects neurons, which contain membrane gangliosides, antigen presentation of these infected cells may trigger the observed autoimmune anti-ganglioside antibodies suggesting direct infection-induced autoantibodies as a cause leading to GBS development. Collectively, our results establish a link between anti-ganglioside antibodies and Zika-associated GBS in patients.

Author summary

Zika virus infection can trigger the development of Guillain Barré syndrome (GBS), a neurological autoimmune disorder mediated by antibodies recognizing gangliosides in nerve membranes. Mechanisms such as molecular mimicry have been identified as a cause for GBS development in certain infections, such as *Campylobacter jejuni*, but the broad self reactivity observed during GBS suggests a role for alternative mechanisms. Our finding that Zika patients with GBS present higher levels of anti-ganglioside antibodies compared to uncomplicated Zika patients in Brazil points to these auto-antibodies as a trigger for GBS in these patients. These findings further support infection-induced autoantibodies as a factor contributing to GBS development, adding novel mechanisms for GBS development beyond molecular mimicry.

Introduction

Zika virus is an arbovirus (arthropod-borne) of the Flaviviridae family, which like dengue viruses and alphavirus chikungunya virus is transmitted by Aedes mosquitoes. Although discovered in 1947 in Uganda, the first large Zika virus outbreak was reported in Micronesia in 2007 [1] followed by the 2014 French Polynesia outbreak [2] and the massive Latin American outbreak in 2015, which was first reported in Brazil and spread across the Americas [3]. During this outbreak, alarming Zika-associated complications, such as microcephaly and Guillain Barré syndrome (GBS) were reported [4, 5].

GBS is an inflammatory neuropathy and the most common cause of neuromuscular paralysis in the world [6]. The etiology of GBS is unknown but its development has been highly associated with post-infection autoimmune responses against gangliosides in peripheral nerves. Gangliosides are sialated glycosphingolipids found in neuronal membranes and are involved in different neuronal functions. Autoimmune antibodies recognizing gangliosides are found in a high proportion of patients with GBS (62% [12]) and are thought to contribute to neuronal pathology inducing complement-mediated axonal injury and demyelination [6]. Molecular mimicry has been proposed as a likely mechanism in infection-induced GBS, where antibodies generated against microbial antigens with structural similarities to specific gangliosides would cross-react with host gangliosides in neuronal membranes. A classic example is GBS associated with *Campylobacter jejuni* infection [7]. Zika virus was added to the list of GBS-associated pathogens due to the high incidence reported during the 2015 Latin America outbreak [8]; however, Zika virus-associated GBS shows anti-gangliosides antibodies (anti-GA1) that cannot be attributed to molecular mimicry [9], as described for *C. jejuni* [7], suggesting alternative mechanisms for the generation of autoantibodies as a result of Zika infection.

During many autoimmune disorders, such as rheumatoid arthritis, autoantibodies play an essential pathological role in mediating the disease. Interestingly, increased levels of IgG autoantibodies against the ganglioside GD3 have been observed in patients with acute Zika infection and without neurologic manifestations such as GBS [10]. Some GBS manifestations have also been associated with elevated levels of autoantibodies such as anti-ganglioside antibodies that can target peripheral nerves [11, 12], but the association of these antibodies with Zikainduced GBS remains unclear.

In this study we evaluate the antibody reactivity levels against 17 different glycolipids, including mostly gangliosides, presented in single and combination form, in the plasma of Zika-infected patients from one of the locations of the 2015 outbreak in Salvador, Brazil. We observed that Zika-associated GBS patients have significantly higher levels of plasma anti-gly-colipid antibodies compared to non-GBS Zika-infected patients. We also observed a broad repertoire of glycolipids, including gangliosides, that were targeted by both IgM and IgG anti-self antibodies. Collectively, these results established a link between anti-ganglioside antibodies and Zika-associated GBS patients.

Methods

Ethics statement

This study was approved by the institutional review board of Instituto Gonçalo Moniz-Fiocruz-n°1184454/2015. All participants were adults, agreed to participate in the study and signed Informed Consent.

Study design and sample collection

Cases of GBS and encephalitis associated with arbovirus infection and Zika infection without neurological symptoms were enrolled in a surveillance study in neurological units of two

reference hospitals in Salvador, Bahia, Brazil, from May 2015 to April 2016, during the Zika outbreak in this area [13]. The study population were patients with acute neurological syndromes admitted to neurology sectors of participating hospitals. Patients with Zika infections but no neurological signs were recruited as part of a surveillance program for Zika infections in the same hospitals. All patients with neurological syndromes were evaluated by the researcher neurologist and the diagnosis of GBS was established according to international criteria [14]. The inclusion criteria were: (1) Patients with symptoms compatible with GBS and its variants or encephalitis. The diagnosis of GBS, Miller-Fisher syndrome (MFS) and its variants [14]; and encephalitis [15] was predetermined by disease-specific criteria. [2] Patients that reported acute exantemathous or fever illness in the 4 weeks before onset of neurologic symptoms. Electromyography and nerve conduction studies were performed in patients with GBS. See Table 1 for details regarding the timing of neurologic symptoms and sample collection in relations to symptoms of arbovirus infection.

Serological analysis

Detection of specific anti-Zika, anti-chikungunya, and anti-dengue IgG antibodies and antidengue and anti-chikungunya IgM antibodies were performed using indirect enzyme-linked

	Patient ID	Zika				GBS	Time to neuro onset (days) ¹	Time to sample collection (days) ²	Chikungunya			Dengue		
		RNA	NT ³	IgM	IgG	Туре			RNA	IgM	IgG	RNA	IgM	IgG
Acute Zika (n = 3)	1	+		+	+	-		2	-	-	-	-	-	+
	2	-		+	-	-		7	-	-	-	-	-	-
	3	+		-	-	-		3	-	-	-	-	-	-
4 months after Zika (n = 2)	4	-		-	+	-		122		-	-		-	-
	5	-		-	+	-		146		-	-		-	+
Zika + encephalitis (n = 3)	6	-		+	+	-	6	7	-	-	+	-	+	+
	7	-		+	+	-	6	5	-	-	-	-	-	-
	8	-		+	+	-	5	13	-	-	-	-	-	+
Zika + GBS (n = 7)	$9(2)^4$	-		+	+	BFP ⁵	10	$10/46^{6}$	-	-	-	-	-	+
	10 (2)	-		+	+	Classic	33	17/57	-	-	-	-	-	+
	11	-		-	+	Classic	4	6	-	-	-	-	-	-
	12	-	+	+	+	Classic	11	5	-	-	-	-	-	+
	13	-	+	+	+	Classic	10	23	-	-	-	-	-	+
	14	-	+	+	+	MFS ⁷	33	5	-	-	-	-	-	+
	15	-		-	+	Classic	3	35	-	-	-	-	-	+
GBS unknown etiology (n = 1)	16	-		-	-	Classic	14	7	-	-	-	-	-	+
Chikungunya (n = 2)	17	-		-	+	Classic	20		-	+	+	-	-	+
	18	-		-	+	Classic	20		-	+	+	-	-	+

T.1.1. 1	D.4!	1.4.4.4	DNIA (L., DT DC	T	-Marill-Char	TT TC A
I adle I.	Patient diagnosis and	detection of Zika	KNA (DV KI-PU	(K) and arbovirus I	givi and igt dv	ELISA.
					A	

¹Days between onset arboviral symptoms (or fever for patient 16) and neurological onset

²Days between onset of arboviral symptoms and sample collection

³Plaque reduction neutralization test

⁴Number of samples collected from an individual patient

⁵Bifacial weakness with paresthesias (BFP)

⁶Days between onset of arboviral symptoms and sample collection for first and second samples

⁷Miller Fisher syndrome (MFS)

https://doi.org/10.1371/journal.pntd.0007695.t001

immunosorbent assays (ELISAs) (Euroimmun, Lüberg, Germany), in accordance with the manufacturer protocol. An IgM antibody-capture ELISA (MAC-ELISA), provided by the Arbovirus Reference Collection division of the Centers for Disease Control and Prevention (CDC), was used in accordance with the established CDC protocol. Detection of RNA for Zika, chikungunya and dengue virus were performed by reverse transcriptase-PCR following published methods [16–18]. Patient samples positive for Zika plaque reduction neutralization test (PRNT) and/or positive for Zika IgM and negative for dengue IgM by ELISA (CDC) were considered positive for Zika infection. Patient 11 was considered positive for Zika infection because it showed positive Zika IgG and negative dengue IgG. Only in patient 15, which is positive for Zika and dengue IgG, lack of cross-reactivity with anti-dengue antibodies could not be confirmed. Zika infection was considered acute when samples were positive for Zika RNA (by RT-PCR, [18] and/or Zika IgM (by ELISA). Biological samples, including blood, were collected upon hospital admission or 4 months after the onset of symptoms, as indicated. Data management was performed using REDCap 6.18.1 - 2018 Vanderbilt University.

Ganglioside ELISAs

Costar 3700 384-well ELISA plates were coated with single or mixes of Glycolipids (Matreya, Sigma) at 20 µg/ml in 200 proof Molecular Biology ethanol using an Agilent Bravo system in a BSL-2 hood. The lipids used were: sphingomyelin (SPM), phosphatidylserine (PS), sulfatide (SULF), globoside (GS), Trihexosylceramide (CTH-hydroxi fatty acid) (THCH), Trihexosylceramide (CTH non-hydroxy fatty acid)(THCHN), galactocerebroside (GALC), and the gangliosides GM1, GM2, GM3, GA1, GD1A, GD1B, GD2, GD3, GT1B and GQ1B. Plates were then allowed to evaporate at RT after >16 h of incubation at 4°C. Plates were washed 3 times with PBS 0.05% tween 20 and then blocked overnight with PBS supplemented with 3% BSA. Plasma from patients was diluted at 1:100 in blocking buffer and incubated for 2 h at 37 °C. Plates were washed again 3 times and incubated with antihuman IgM/IgG-HRP (Abcam) for 1 h at 37 °C. Plates were washed 3 more times and TMB substrate (BD Biosciences) was added until desired color was obtained. Reaction was stopped with Stop buffer (Biolegend) and absorbance was read at 450 nm. The optical density at 450 nm was compared with the same dilution (1:100) of a positive plasma sample (sample ID: 9b) that was used as reference to calculate relative units (RU). Two negative controls were included in each of the ELISA plates run: (1) The plasma of a healthy US control donor was used in duplicated wells in the ELISA for each glycolipid and combinations. The average of the 2 determinations for each glycolipid and combinations was used as background for each glycolipid and subtracted from each value. (2) The reactivity of each plasma sample in wells coated with PBS supplemented with 3% BSA. The value of eight independent wells for each plasma sample was obtained. It was observed that the variation between the eight replicates with each plasma samples was <0.02 for all samples and the variation between the average values for the different plasma samples was <0.0002. The reactivity of all plasma samples to BSA was considered constant and was not subtracted from assay values. The reactivity to wells coated with only glycolipids coating buffer (ethanol) was not considered since it results in high unspecific background for all samples. The secondary antibody, TMB and stop solution was added using the peristaltic pump on the Biotek EL406. Washes were also done using the 96-head washer on the EL406. Validation of the automated 384-well ELISAs was performed using a similar protocol in 96-well plates with manual pipetting. Five different plasma samples (9b, 10a, 11, 14 and 16) of GBS patients were tested with five randomly chosen glycolipids (SPM, GS, GM1, GD3 and SULF). The variation between the two assays was found to be lower than 0.008 for each of the glycolipids.

Statistical analysis

Data were analyzed using Prism (GraphPad Software). Unpaired t-test was used to identify statistical differences between groups of samples. For determination of number of antigens recognized per sample and number of positive samples recognizing each ganglioside, reactivity of samples was considered positive if the OD value was at least the average of control background wells plus three times the standard deviation.

Results

Zika-associated GBS is one of the most serious complications associated with this infection. Our main goal in this study is to compare the prevalence and specificities of anti-ganglioside antibodies in a cohort of Zika-associated GBS patients (n = 7) compared to Zika-infected patients without GBS (n = 8). Among these, some patients presented with encephalitis (n = 3) or were Zika-infected with no neurological symptoms, either in acute phase (n = 3) or 4 months after the onset of symptoms (n = 2). The median of days between the onset of arboviral and neurological symptoms was 10 days for the Zika-associated GBS patients (Table 1).

Strong anti-ganglioside antibody response in Zika-associated GBS patients

Using a novel high throughput ELISA approach, we assessed the plasma of these patients for their reactivity (IgM and IgG) against different gangliosides. The reactivity to combinations of glycolipids has been described to be higher than the reactivity to single ones, possibly due to the formation of complex antigenic structures [12]. We therefore analyzed reactivity against 17 lipids, mostly glycolipids including gangliosides described to be associated with GBS [6], alone or their 139 double combinations. We first performed an overall analysis of all the patient samples to determine the levels of ganglioside reactivity in their plasma using a high throughput ELISA approach to detect anti-ganglioside IgM and IgG antibodies. These assays showed increased anti-ganglioside reactivity in the plasma of Zika-associated GBS patients compared to Zika patients without GBS (Fig 1A–1C). Collectively, these results showed an enriched anti-ganglioside antibody response in the plasma of Zika-associated GBS patients compared to Zika-infected controls.

Additionally, we analyzed one sample from a GBS patient of unknown etiology who was negative for Zika, dengue and chikungunya infections, which showed a similar profile to Zika-associated GBS patients.

Since a previous report described that the plasma of GBS patients presented higher antibody reactivity to complex glycolipids compared to individual ones, we analyzed the responses to individual versus 2-by-2 combined glycolipids. We did not find any significant differences between the average reactivity of any of the plasma samples to individual or combined glycolipids (Fig 2).

We also analyzed the plasma of two GBS patients with active chikungunya virus infection (IgM+) and previous Zika infection (IgG+). The plasma from these patients did not present strong reactivity against gangliosides, in contrast to Zika-associated GBS patients (Fig 3).

Broad ganglioside specificity in the plasma of Zika-associated GBS patients

We further dissected the anti-ganglioside reactivity patterns observed in Zika-infected patients with or without GBS. A detailed analysis of antigen reactivity showed that patients that developed GBS recognized a significantly higher number of antigens (single and combined ganglioside mixes) compared to patients without GBS (Fig 4A). The highest number of antigens recognized was found in a GBS patient reacting significantly to >100 different single



Fig 1. Reactivity of plasma samples from Zika patients with or without GBS. (a) Heatmap of antibody reactivity to single lipids and lipid combinations. Results from plasma of patients with uncomplicated Zika (acute (I: patients 1–3) or 4 months after the onset of symptoms (II: patients 4–5), with Zika infection and encephalitis (III: patients 6–8), with Zika infection and GBS (patients 9–15; a and b indicate different time points of sampling for the same patient) or with GBS from unidentified etiology (patient 16). Antigen lipids used: sphingomyelin (SPM), phosphatidylserine (PS), sulfatide (SULF), globoside (GS), Trihexosylceramide (CTH-hydroxi fatty acid) (THCH), Trihexosylceramide (CTH non-hydroxy fatty acid)(THCHN), galactocerebroside (GALC), and the gangliosides GM1, GM2, GM3, GA1, GD1A, GD1B, GD2, GD3, GT1B and GQ1B. (b) Average of reactivity values for all lipids and combinations in samples from patients with Zika with or without GBS. Samples of plasma from a zika infection (blue circles), Zika with encephalitis (red circles) and Zika with GBS (black squares). A sample of plasma from a patient with GBS but no Zika infection (unidentified etiology) (green square) was not considered in the average. (c) Average of reactivity values for each lipid and lipid

combinations in samples from patients with Zika with or without GBS. Data are expressed as RU values. *p < 0.05, **p < 0.01, ***p < 0.001, ****p < 0.001, when the groups are compared to each other by unpaired t-test.

https://doi.org/10.1371/journal.pntd.0007695.g001

gangliosides/combinations while all non-GBS patients reacted significantly against 2 or fewer antigens. When we analyzed which specific gangliosides had an enriched reactivity across the Zika infected patients (either in single or combination form) we observed a broad reactivity to different gangliosides and other glycolipids (Fig 4B). These results suggest a broad anti-ganglioside antibody response in Zika-associated GBS patients, independent of combinations.

Isotypes of antibodies in the anti-ganglioside response in Zika-associated GBS high responders

Our initial screen of anti-ganglioside reactivity evaluated the presence of overall IgM/IgG antibodies in the plasma of the Zika-associated GBS patients. We further analyzed the samples by validating the responses observed in six highly responsive patients, from the Zika-associated GBS group. We also dissected this response by determining IgM and IgG reactivities separately (Fig 5). The results of these assays showed high levels of both anti-ganglioside IgM and IgG in the plasma of the patients compared to control plasma. Additionally, these assays validated the strong broad reactivity of the Zika-GBS patient plasma against a high number of gangliosides (Fig 4). Collectively, these results confirm the presence of IgM and IgG anti-ganglioside antibodies in the plasma of Zika-associated GBS patients. Our findings demonstrate broad reactivity to glycolipids with stronger responses to specific gangliosides.

Discussion

GBS is one of the most serious complications associated with Zika infection. Neurological symptoms appear shortly after a transient Zika infection resulting in the development of GBS [9, 11, 19, 20]. In this study a median of 10 days before the onset of neurological symptoms



Fig 2. Reactivity to single glycolipids versus combinations. Average antibody reactivity of samples from patients with Zika or Zika + GBS was analyzed for single and combined glycolipids. Each dot represents the average reactivity to all single or combined glycolipids for each plasma sample. No significant (ns) differences between single and combined groups were found when compared by unpaired t-test.

https://doi.org/10.1371/journal.pntd.0007695.g002



Fig 3. Reactivity of samples from plasma of two GBS patients with active chikungunya virus infection and previous Zika infection (IgG+). (a) Heatmap of plasma reactivity to single lipids and lipid combinations. Results from plasma from patients with active chikungunya virus infection (IgM+) and previous Zika infection (IgG+) and GBS compared with Zika infection alone and GBS. (b) Average of reactivity values for all lipids and combinations in samples from these patients. (c) Average of reactivity values for each lipid and lipid combinations in samples from these patients. Data are expressed as RU values. *p < 0.05, **p < 0.01, ***p < 0.001, ****p < 0.0001, when the groups are compared to each other by unpaired t-test.

https://doi.org/10.1371/journal.pntd.0007695.g003

was observed, which is similar to a previous report (6 days) [9]. Although Zika virus infection has been reported to lead to development of GBS in patients, little is known about the pathogenesis of this syndrome. Different autoantibodies have been identified as mediators of pathology during different autoimmune disorders, such as anti-nuclear antibodies during Systemic Lupus Erythematosus [21]. Indeed, GBS is considered an autoimmune syndrome due to the immune destruction of peripheral nerve components such as gangliosides [19, 22, 23]. Generation of anti-ganglioside antibodies and other autoantibodies has been reported in non-Zika infection-induced GBS, such as the classical one induced by the bacterium *C. jejuni* [7]. Because anti-ganglioside autoantibodies are implicated in the pathogenesis of GBS [6], we sought to determine whether anti-ganglioside antibodies were selectively increased in Zika-infected patients with GBS as opposed to Zika-infected controls with self-limited illness.

Our initials results show a potent broad reactivity against single and combination of 17 different gangliosides compared to non-GBS Zika infected patients. These anti-ganglioside



Antigens recognized

Fig 4. Antigenic analysis of the reactivity of plasma from Zika patients with or without GBS. (a) Reactivity to glycolipids, presented single or as combinations, of the plasma samples from Zika patients (identified from 1 to 15; a and b indicate different time points of sampling for the same patient). The patient with GBS of unknown etiology (16) is marked with a red rectangle. Data is presented as number of single lipids or combinations that were recognized by plasma samples of each patient. (b) Reactivity to single glycolipids in plasma samples of Zika-associated GBS patients (n = 9). A sample was considered positive if its value was higher than the average plus 3 three times the standard deviation of the values obtained for the eight Zika without GBS patient samples.

https://doi.org/10.1371/journal.pntd.0007695.g004

antibodies were both of the IgM and IgG isotypes. Both anti-ganglioside IgG and IgM have been suggested to have a pathological role during different non-Zika infection induced GBS patients [24]. The mechanism by which these anti-gangliosides lead to pathology is poorly understood. The significant increase in anti-ganglioside IgM/IgG antibodies in patients with GBS compared to non-GBS Zika-infected patients suggests a role for these antibodies in mediating the disease.

A recent study assessed a similar relationship of ganglioside reactivity in Zika-associated GBS patients in French Polynesia [9]. This study found an increase in general anti-ganglioside reactivity against a different set of gangliosides tested by a different method of combinatorial glycolipid microarray. Our results show reactivity against different individual and combination of gangliosides hence illustrating the diversity in the anti-ganglioside response induced in



Fig 5. IgM and IgG reactivity of plasma from high-responder Zika patients with GBS. Heatmap of plasma IgM and IgG reactivity to individual glycolipids, expressed as optical density values from the specific ELISAs. The patient with GBS of unknown etiology (16) is marked with a red rectangle. Control uninfected plasma (C).

https://doi.org/10.1371/journal.pntd.0007695.g005

a different cohort of Zika-associated GBS patients. We did not observe any enhancement of reactivity whenever specific gangliosides were tested individually or in combination, which may be attributed to the different methods used for detection (microarray versus ELISA).

In addition to Zika virus, other arbovirus infections like dengue and chikungunya have also been reported to lead to autoimmunity and neurological problems such as GBS [25–30]. A large percentage of the patients in our study had previous, but not active, dengue infections, as indicated by the differential anti-dengue IgM and IgG reactivity. Previous dengue infections are expected in this area of Brazil where prevalence is 86% in adults and where preexisting high antibody titers to dengue virus have been associated with reduced risk of Zika infection [31].

We also assessed the plasma of two GBS patients with active chikungunya virus infection and previous Zika infection (IgG+). When we assessed the plasma of these patients for antiganglioside reactivity, our results demonstrated levels of anti-ganglioside antibodies significantly lower than GBS patients with an active Zika infection. It is possible that anti-ganglioside antibodies in the circulation induced during a Zika infection decrease over time and are no longer present in these patients. The role of an active chikungunya infection is unclear.

It is well established that Zika can infect neurons [32–34], including peripheral motor nerves/nerve roots, which have high abundance of different gangliosides. Direct infection of neurons would target these cells for phagocytosis by antigen presenting cells, enabling presentation of many auto antigens, such as gangliosides, along with virus antigens, resulting in an antibody response against both. Increased immune recognition of virus-infected neuronal cell antigens could be at the basis of Zika–induced GBS. Immune response against direct neural infection would be consistent with the observation that GBS tends to be an early complication of Zika. Accordingly, a recent studied showed how antiviral CD8+ T-cells mediated nerve damage leading to paralysis in Zika-infected mice [35]. Additionally, immune mediated

neurological damage was also reported in fatal cases of Zika-induced microcephaly [36], providing additional evidence of an immune component contributing to the neuronal damage leading to GBS. Accordingly, plasma from Brazilian Zika-infected patients recognized GD3 from neurons in retina tissues [10]. These patients were found to have high titers of anti-ganglioside antibodies, mainly anti-GD3 IgG antibodies. Nevertheless, mechanistic studies are needed to test this hypothesis and elucidate the role anti-ganglioside antibodies might have in Zika-induced GBS.

Although we observed wide reactivity to gangliosides, our results also showed differential reactivity to some gangliosides in the plasma of Zika-associated GBS patients. When we dissected these responses by validating them with single ganglioside ELISAs, we confirmed a strong reactivity against specific gangliosides such as GA1. Interestingly, GA1 also had the highest reactivity from the French Polynesia study as assessed by a different method of combinatorial glycolipid microarray [9]. However, in a different study in India with Zika-infected GBS patients, the most commonly recognize ganglioside was GT1b [37]. Collectively, our results suggest that, in a minor subset of infected patients, Zika infection causes neuronal damage that triggers an auto-immune antibody response against neuron-derived ganglioside antigens, which contributes to the pathogenesis of GBS.

Acknowledgments

We thank the High-throughput Biology Laboratory Core services at NYU Langone Medical Center.

Author Contributions

Conceptualization: Juan Rivera-Correa, Isadora Cristina de Siqueira, Joel D. Ernst, Ana Rodriguez.

Data curation: Juan Rivera-Correa, Ana Rodriguez.

Formal analysis: Juan Rivera-Correa, Ana Rodriguez.

Funding acquisition: Joel D. Ernst, Ana Rodriguez.

Investigation: Sabrina Mota.

Methodology: Juan Rivera-Correa, Joel D. Ernst.

Project administration: Joel D. Ernst.

Resources: Isadora Cristina de Siqueira, Mateus Santana do Rosário, Pedro Antônio Pereira de Jesus, Luiz Carlos Junior Alcantara, Joel D. Ernst, Ana Rodriguez.

Supervision: Joel D. Ernst, Ana Rodriguez.

Writing - original draft: Juan Rivera-Correa.

Writing - review & editing: Isadora Cristina de Siqueira, Joel D. Ernst, Ana Rodriguez.

References

- Duffy MR, Chen TH, Hancock WT, Powers AM, Kool JL, Lanciotti RS, et al. Zika virus outbreak on Yap Island, Federated States of Micronesia. N Engl J Med. 2009; 360(24):2536–43. https://doi.org/10.1056/ NEJMoa0805715 PMID: 19516034
- Cao-Lormeau VM, Roche C, Teissier A, Robin E, Berry AL, Mallet HP, et al. Zika virus, French polynesia, South pacific, 2013. Emerg Infect Dis. 2014; 20(6):1085–6. https://doi.org/10.3201/eid2006.140138 PMID: 24856001

- Ikejezie J, Shapiro CN, Kim J, Chiu M, Almiron M, Ugarte C, et al. Zika Virus Transmission—Region of the Americas, May 15, 2015-December 15, 2016. MMWR Morb Mortal Wkly Rep. 2017; 66(12):329– 34. https://doi.org/10.15585/mmwr.mm6612a4 PMID: 28358795
- Dirlikov E, Torres JV, Martines RB, Reagan-Steiner S, Perez GV, Rivera A, et al. Postmortem Findings in Patient with Guillain-Barre syndrome and Zika Virus Infection. Emerg Infect Dis. 2018; 24(1):114–7.
- Hoen B, Schaub B, Funk AL, Ardillon V, Boullard M, Cabie A, et al. Pregnancy Outcomes after ZIKV Infection in French Territories in the Americas. N Engl J Med. 2018; 378(11):985–94. <u>https://doi.org/10.1056/NEJMoa1709481</u> PMID: 29539287
- Goodfellow JA, Willison HJ. Guillain-Barre syndrome: a century of progress. Nat Rev Neurol. 2016; 12 (12):723–31. https://doi.org/10.1038/nrneurol.2016.172 PMID: 27857121
- Godschalk PC, Kuijf ML, Li J, St Michael F, Ang CW, Jacobs BC, et al. Structural characterization of Campylobacter jejuni lipooligosaccharide outer cores associated with Guillain-Barre and Miller Fisher syndromes. Infect Immun. 2007; 75(3):1245–54. https://doi.org/10.1128/IAI.00872-06 PMID: 17261613
- Sebastian UU, Ricardo AVA, Alvarez BC, Cubides A, Luna AF, Arroyo-Parejo M, et al. Zika virusinduced neurological critical illness in Latin America: Severe Guillain-Barre syndrome and encephalitis. J Crit Care. 2017; 42:275–81. https://doi.org/10.1016/j.jcrc.2017.07.038 PMID: 28806562
- Cao-Lormeau VM, Blake A, Mons S, Lastere S, Roche C, Vanhomwegen J, et al. Guillain-Barre syndrome outbreak associated with Zika virus infection in French Polynesia: a case-control study. Lancet. 2016; 387(10027):1531–9. https://doi.org/10.1016/S0140-6736(16)00562-6 PMID: 26948433
- Nico D, Conde L, Rivera-Correa JL, Vasconcelos-Dos-Santos A, Mesentier-Louro L, Freire-de-Lima L, et al. Prevalence of IgG Autoantibodies against GD3 Ganglioside in Acute Zika Virus Infection. Front Med (Lausanne). 2018; 5:25.
- Pinto-Diaz CA, Rodriguez Y, Monsalve DM, Acosta-Ampudia Y, Molano-Gonzalez N, Anaya JM, et al. Autoimmunity in Guillain-Barre syndrome associated with Zika virus infection and beyond. Autoimmun Rev. 2017; 16(4):327–34. https://doi.org/10.1016/j.autrev.2017.02.002 PMID: 28216074
- Rinaldi S, Brennan KM, Kalna G, Walgaard C, van Doorn P, Jacobs BC, et al. Antibodies to heteromeric glycolipid complexes in Guillain-Barre syndrome. PLoS One. 2013; 8(12):e82337. <u>https://doi.org/10.1371/journal.pone.0082337 PMID: 24358172</u>
- do Rosário M, Jesus P, Farias D, Novaes MA, Moura D, Lima F, et al. Neurological complications associated with arboviruses during Zika outbreak in Salvador, Bahia-Brazil. International Journal of Infectious Diseases; 2018; Suppl. 73:49. https://doi.org/10.1016/j.ijid.2018.05.016
- Wakerley BR, Uncini A, Yuki N. Guillain-Barre and Miller Fisher syndromes—new diagnostic classification. Nat Rev Neurol. 2014; 10(9):537–44. https://doi.org/10.1038/nrneurol.2014.138 PMID: 25072194
- Steiner I, Budka H, Chaudhuri A, Koskiniemi M, Sainio K, Salonen O, et al. Viral encephalitis: a review of diagnostic methods and guidelines for management. Eur J Neurol. 2005; 12(5):331–43. <u>https://doi.org/10.1111/j.1468-1331.2005.01126.x PMID: 15804262</u>
- Johnson BW, Russell BJ, Lanciotti RS. Serotype-specific detection of dengue viruses in a fourplex realtime reverse transcriptase PCR assay. J Clin Microbiol. 2005; 43(10):4977–83. https://doi.org/10.1128/ JCM.43.10.4977-4983.2005 PMID: 16207951
- Lanciotti RS, Kosoy OL, Laven JJ, Panella AJ, Velez JO, Lambert AJ, et al. Chikungunya virus in US travelers returning from India, 2006. Emerg Infect Dis. 2007; 13(5):764–7. <u>https://doi.org/10.3201/eid1305.070015</u> PMID: 17553261
- Lanciotti RS, Kosoy OL, Laven JJ, Velez JO, Lambert AJ, Johnson AJ, et al. Genetic and serologic properties of Zika virus associated with an epidemic, Yap State, Micronesia, 2007. Emerg Infect Dis. 2008; 14(8):1232–9. https://doi.org/10.3201/eid1408.080287 PMID: 18680646
- Anaya JM, Ramirez-Santana C, Salgado-Castaneda I, Chang C, Ansari A, Gershwin ME. Zika virus and neurologic autoimmunity: the putative role of gangliosides. BMC Med. 2016; 14:49. <u>https://doi.org/ 10.1186/s12916-016-0601-y</u> PMID: 27001187
- Monsalve DM, Pacheco Y, Acosta-Ampudia Y, Rodriguez Y, Ramirez-Santana C, Anaya JM. Zika virus and autoimmunity. One-step forward. Autoimmun Rev. 2017. https://doi.org/10.1016/j.autrev.2017.10. 008 PMID: 29037898
- Leuchten N, Hoyer A, Brinks R, Schoels M, Schneider M, Smolen J, et al. Performance of Antinuclear Antibodies for Classifying Systemic Lupus Erythematosus: A Systematic Literature Review and Meta-Regression of Diagnostic Data. Arthritis Care Res (Hoboken). 2018; 70(3):428–38.
- Ravindranath MH, Muthugounder S, Saravanan TS, Presser N, Morton DL. Human antiganglioside autoantibodies: validation of ELISA. Ann N Y Acad Sci. 2005; 1050:229–42. https://doi.org/10.1196/ annals.1313.024 PMID: 16014538

- 23. van der Meche FG, van Doorn PA. Guillain-Barre syndrome and chronic inflammatory demyelinating polyneuropathy: immune mechanisms and update on current therapies. Ann Neurol. 1995; 37 Suppl 1: S14–31. PMID: 8968214
- Koga M, Takahashi M, Yokoyama K, Kanda T. Ambiguous value of anti-ganglioside IgM autoantibodies in Guillain-Barre syndrome and its variants. J Neurol. 2015; 262(8):1954–60. https://doi.org/10.1007/ s00415-015-7806-4 PMID: 26050638
- Arroyo-Avila M, Caban A, Garcia-Rivera EJ, Irizarry-Perez M, Torres H, Gorbea H, et al. Clinical Manifestations Associated with Peripheral Joint Involvement in Patients with Acute Chikungunya Virus Infection. Am J Trop Med Hyg. 2017; 96(4):916–21. https://doi.org/10.4269/ajtmh.16-0890 PMID: 28138049
- Flodstrom-Tullberg M. Viral infections: their elusive role in regulating susceptibility to autoimmune disease. Microbes Infect. 2003; 5(10):911–21. PMID: 12919859
- Manimunda SP, Vijayachari P, Uppoor R, Sugunan AP, Singh SS, Rai SK, et al. Clinical progression of chikungunya fever during acute and chronic arthritic stages and the changes in joint morphology as revealed by imaging. Trans R Soc Trop Med Hyg. 2010; 104(6):392–9. <u>https://doi.org/10.1016/j.trstmh.</u> 2010.01.011 PMID: 20171708
- Mehta R, Soares CN, Medialdea-Carrera R, Ellul M, da Silva MTT, Rosala-Hallas A, et al. The spectrum of neurological disease associated with Zika and chikungunya viruses in adults in Rio de Janeiro, Brazil: A case series. PLoS Negl Trop Dis. 2018; 12(2):e0006212. <u>https://doi.org/10.1371/journal.pntd.</u> 0006212 PMID: 29432457
- Tanay A. Chikungunya virus and autoimmunity. Curr Opin Rheumatol. 2017; 29(4):389–93. <u>https://doi.org/10.1097/BOR.00000000000396</u> PMID: 28376065
- Wan SW, Lin CF, Yeh TM, Liu CC, Liu HS, Wang S, et al. Autoimmunity in dengue pathogenesis. J Formos Med Assoc. 2013; 112(1):3–11. https://doi.org/10.1016/j.jfma.2012.11.006 PMID: 23332423
- Rodriguez-Barraquer I, Costa F, Nascimento EJM, Nery NJ, Castanha PMS, Sacramento GA, et al. Impact of preexisting dengue immunity on Zika virus emergence in a dengue endemic region. Science. 2019; 363(6427):607–10. https://doi.org/10.1126/science.aav6618 PMID: 30733412
- Gabriel E, Ramani A, Karow U, Gottardo M, Natarajan K, Gooi LM, et al. Recent Zika Virus Isolates Induce Premature Differentiation of Neural Progenitors in Human Brain Organoids. Cell Stem Cell. 2017; 20(3):397–406 e5. https://doi.org/10.1016/j.stem.2016.12.005 PMID: 28132835
- Li C, Xu D, Ye Q, Hong S, Jiang Y, Liu X, et al. Zika Virus Disrupts Neural Progenitor Development and Leads to Microcephaly in Mice. Cell Stem Cell. 2016; 19(5):672. <u>https://doi.org/10.1016/j.stem.2016.10.</u> 017 PMID: 27814481
- Tang H, Hammack C, Ogden SC, Wen Z, Qian X, Li Y, et al. Zika Virus Infects Human Cortical Neural Progenitors and Attenuates Their Growth. Cell Stem Cell. 2016; 18(5):587–90. <u>https://doi.org/10.1016/j.stem.2016.02.016</u> PMID: 26952870
- Jurado KA, Yockey LJ, Wong PW, Lee S, Huttner AJ, Iwasaki A. Antiviral CD8 T cells induce Zika-virusassociated paralysis in mice. Nat Microbiol. 2018; 3(2):141–7. <u>https://doi.org/10.1038/s41564-017-</u> 0060-z PMID: 29158604
- 36. Azevedo RSS, de Sousa JR, Araujo MTF, Martins Filho AJ, de Alcantara BN, Araujo FMC, et al. In situ immune response and mechanisms of cell damage in central nervous system of fatal cases microcephaly by Zika virus. Sci Rep. 2018; 8(1):1. https://doi.org/10.1038/s41598-017-17765-5 PMID: 29311619
- Baskar D, Amalnath D, Mandal J, Dhodapkar R, Vanathi K. Antibodies to Zika virus, Campylobacter jejuni and gangliosides in Guillain-Barre syndrome: A prospective single-center study from southern India. Neurol India. 2018; 66(5):1324–31. https://doi.org/10.4103/0028-3886.241402 PMID: 30232998